Author’s response to reviews

Title: Parameters of ovarian reserve in relation to urinary concentrations of parabens.

Authors:
Joanna Jurewicz (joanna.jurewicz@imp.lodz.pl)
Michał Radwan (mrad@gameta.pl)
Bartosz Wielgomas (bart@gumed.edu.pl)
Anetta Karwacka (jak1972@wp.pl)
Anna Klimowska (ann@gumed.edu.pl)
Pawel Kaluzny (jg-wp@wp.pl)
Pawel Radwan (prad@gameta.pl)
Wojciech Hanke (jona278@wp.pl)

Version: 4 Date: 08 Jan 2020

Author’s response to reviews:

Answer to Reviewers:

ENHE-D-19-00047R3
Parameters of ovarian reserve in relation to urinary concentrations of parabens.
Joanna Jurewicz; Michał Radwan; Bartosz Wielgomas; Anetta Karwacka; Anna Klimowska;
Pawel Kaluzny; Pawel Radwan; Wojciech Hanke
Environmental Health

Technical Comments from the Editorial Office:
If you could please go through your manuscript one final time and do some light copy editing, we think that would help to move the manuscript forward to an acceptance.

Answer: Some light copy editing have been performed in the whole manuscript.